%~MoulinÃ© par MaN_auto v.0.34.0 (b272df1a) 2025-06-03 12:13:30
\documentclass[CRMATH,Unicode,XML,biblatex]{cedram}

\TopicFR{Analyse et gÃ©omÃ©trie complexes}
\TopicEN{Complex analysis and geometry}

\addbibresource{CRMATH_Ma_20250141.bib}

\usepackage{mathtools}

\newcommand{\C}{\mathbb{C}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\D}{\mathbb{D}}

\DeclareMathOperator{\rota}{rot}
\DeclareMathOperator{\gr}{grad}
\DeclareMathOperator*{\Imag}{Im}
\DeclareMathOperator*{\Ric}{Ric}
\DeclareMathOperator*{\PSL}{PSL}

\newcommand{\bL}{\boldsymbol{L}}

\newcommand*{\dd}{\mathrm{d}}
\newcommand*{\Ber}{\mathrm{Ber}}
\newcommand*{\FS}{\mathrm{FS}}

%%% \relabel
\newcommand*{\relabel}{\renewcommand{\labelenumi}{(\theenumi)}}

%%% Change counter appearance
\newcommand*{\romanenumi}{\renewcommand*{\theenumi}{\roman{enumi}}\relabel}

%%%---------------------------------------------------------------------------

%%% ARROWS

%%% \to macro
\renewcommand*{\to}{\mathchoice{\longrightarrow}{\rightarrow}{\rightarrow}{\rightarrow}}
%% Remplacer \xrightarrow{#1} par \overset{#1}{\longrightarrow}.

%%% \mapsto macro
\let\oldmapsto\mapsto
\renewcommand*{\mapsto}{\mathchoice{\longmapsto}{\oldmapsto}{\oldmapsto}{\oldmapsto}}

%%% \longhookrightarrow macro
%\newcommand{\longhookrightarrow}{\!\lhook\joinrel\longrightarrow}

%%% \longrightrightarrows macro
%\newcommand{\longrightrightarrows}{\overset{\raisebox{-.17ex}{$\displaystyle{\longrightarrow}$}}{\smash{\raisebox{-.3ex}{$\longrightarrow$}}}}

%%% Convergence
%\newcommand\dconverge[2][p \to + \infty]{\xrightarrow[#1]{#2}}
%\newcommand\tconverge[1][p \to + \infty]{\underset{#1}{\longrightarrow}}

%%% Isomorphism
%\newcommand{\isomto}{\overset{\!\!\raisebox{.3ex}{$\scriptstyle{\sim}$}}{\smash{\to}}}

%%%---------------------------------------------------------------------------

%%% BAR & OVERLINE

%%% Better widebar macro
\makeatletter
\let\save@mathaccent\mathaccent
\newcommand*\if@single[3]{%
 \setbox0\hbox{${\mathaccent"0362{#1}}^H$}%
 \setbox2\hbox{${\mathaccent"0362{\kern0pt#1}}^H$}%
 \ifdim\ht0=\ht2 #3\else #2\fi
 }
%The bar will be moved to the right by a half of \macc@kerna, which is computed by amsmath:
\newcommand*\rel@kern[1]{\kern#1\dimexpr\macc@kerna}
%If there's a superscript following the bar, then no negative kern may follow the bar;
%an additional {} makes sure that the superscript is high enough in this case:
\newcommand*\widebar[1]{\@ifnextchar^{{\wide@bar{#1}{0}}}{\wide@bar{#1}{1}}}
%Use a separate algorithm for single symbols:
\newcommand*\wide@bar[2]{\if@single{#1}{\wide@bar@{#1}{#2}{1}}{\wide@bar@{#1}{#2}{2}}}
\newcommand*\wide@bar@[3]{%
 \begingroup
 \def\mathaccent##1##2{%
%Enable nesting of accents:
  \let\mathaccent\save@mathaccent
%If there's more than a single symbol, use the first character instead (see below):
  \if#32 \let\macc@nucleus\first@char \fi
%Determine the italic correction:
  \setbox\z@\hbox{$\macc@style{\macc@nucleus}_{}$}%
  \setbox\tw@\hbox{$\macc@style{\macc@nucleus}{}_{}$}%
  \dimen@\wd\tw@
  \advance\dimen@-\wd\z@
%Now \dimen@ is the italic correction of the symbol.
  \divide\dimen@ 3
  \@tempdima\wd\tw@
  \advance\@tempdima-\scriptspace
%Now \@tempdima is the width of the symbol.
  \divide\@tempdima 10
  \advance\dimen@-\@tempdima
%Now \dimen@ = (italic correction / 3) - (Breite / 10)
  \ifdim\dimen@>\z@ \dimen@0pt\fi
%The bar will be shortened in the case \dimen@<0 !
  \rel@kern{0.6}\kern-\dimen@
  \if#31
   \overline{\rel@kern{-0.6}\kern\dimen@\macc@nucleus\rel@kern{0.4}\kern\dimen@}%
   \advance\dimen@0.4\dimexpr\macc@kerna
%Place the combined final kern (-\dimen@) if it is >0 or if a superscript follows:
   \let\final@kern#2%
   \ifdim\dimen@<\z@ \let\final@kern1\fi
   \if\final@kern1 \kern-\dimen@\fi
  \else
   \overline{\rel@kern{-0.6}\kern\dimen@#1}%
  \fi
 }%
 \macc@depth\@ne
 \let\math@bgroup\@empty \let\math@egroup\macc@set@skewchar
 \mathsurround\z@ \frozen@everymath{\mathgroup\macc@group\relax}%
 \macc@set@skewchar\relax
 \let\mathaccentV\macc@nested@a
%The following initialises \macc@kerna and calls \mathaccent:
 \if#31
  \macc@nested@a\relax111{#1}%
 \else
%If the argument consists of more than one symbol, and if the first token is
%a letter, use that letter for the computations:
  \def\gobble@till@marker##1\endmarker{}%
  \futurelet\first@char\gobble@till@marker#1\endmarker
  \ifcat\noexpand\first@char A\else
   \def\first@char{}%
  \fi
  \macc@nested@a\relax111{\first@char}%
 \fi
 \endgroup
}
\makeatother

\let\oldbar\bar
\renewcommand*{\bar}[1]{{\mathchoice{\widebar{#1}}{\widebar{#1}}{\widebar{#1}}{\oldbar{#1}}}}

% /!\ Il y a parfois des soucis avec cette macro lorsqu'il y a des caractÃ¨res en minuscule en subscript (et dans les captions). Dans ce cas, faire avec \overline et des espaces manuels.

%%% Double bar
% \overline{\overline{\!a}}

%\newcommand\bx{\,\overline{\!x}}
%\newcommand\brho{\,\overline{\!\rho}}
%better widebar a apparemment des soucis avec les \subcaptions pour les CR :shrug:

%%%---------------------------------------------------------------------------

%%% DELIMITERS

%\usepackage{mathtools}
\DeclarePairedDelimiter{\parens}{\lparen}{\rparen}
\DeclarePairedDelimiter{\braces}{\{}{\}}
\DeclarePairedDelimiter{\abs}{\lvert}{\rvert}
\DeclarePairedDelimiter{\norm}{\lVert}{\rVert}
\DeclarePairedDelimiter{\bracks}{[}{]}
\DeclarePairedDelimiter{\bbracks}{\llbracket}{\rrbracket}
\DeclarePairedDelimiter{\angles}{\langle}{\rangle}
\DeclarePairedDelimiter{\aangles}{\langle\!\langle}{\rangle\!\rangle}
\DeclarePairedDelimiter{\pparens}{(\!(}{)\!)}

%%%---------------------------------------------------------------------------

%%% VERTICAL BAR

%%% ''Evaluated at'' macro
\newcommand\restr[2]{{				% we make the whole thing an ordinary symbol
 \left.\kern-\nulldelimiterspace 	% automatically resize the bar with \right
 #1 									% the function
 \vphantom{\big\vert} 				% pretend it's a little taller at normal size
 \right\vert_{#2} 					% this is the delimiter
 }}
% As a subscript: \phi_{F_f\rvert_{K_\ell}} [NO : \phi_{\left.h\right\vert_{K_\ell}}].
% In textstyle : \bigr\rvert_{}.

%%% ''Such that'' macro
\newcommand*{\st}[1][]{\;#1\vert\;}
%\newcommand*{\st}{\;\vert\;}
%\newcommand*{\bigst}{\;\big\vert\;}
%\newcommand*{\Bigst}{\;\Big\vert\;}
%\newcommand*{\biggst}{\;\bigg\vert\;}
%\newcommand*{\Biggst}{\;\Bigg\vert\;}
% Scaling version (with left and right delimiters): \,\middle\vert\,

%%% Divides
% Use \,\big\vert\,
% Or if \nmid is also present (?):
%\newcommand{\zerodel}{.\kern-\nulldelimiterspace}
%\newcommand{\divides}[2]{\left\zerodel #1 \ \middle\vert \ #2 \right\zerodel}
%\newcommand{\divides}[3][\middle]{\left\zerodel #2 \,#1\vert\, #3 \right\zerodel}

%%% Separator (non-scaling)
% \mid

%%%---------------------------------------------------------------------------

\graphicspath{{./figures/}}

\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}

%%%---------------------------------------------------------------------------

%%% Changer la largeur des adresses

% Jouer avec \hsize :
\makeatletter
\def\@setafterauthor{%
  \vglue3mm%
%  \hspace*{0pt}%
\begingroup\hsize=11.95cm\advance\hsize\abstractmarginL\raggedright
\noindent
%\hspace*{\abstractmarginL}\begin{minipage}[t]{10cm}
   \leftskip\abstractmarginL
  \normalfont\Small
  \@afterauthor\par
\endgroup
\vskip2pt plus 3pt minus 1pt
}
\makeatother

%%%---------------------------------------------------------------------------

\title{Fubini--Study forms on punctured Riemann surfaces}
\alttitle{Formes de Fubini--Study sur des surfaces de Riemann \'epoint\'ees}

\author{\firstname{Razvan} \lastname{Apredoaei}}
\address{Universit\'e Paris Cit\'e, CNRS, IMJ-PRG, B\^atiment Sophie Germain, UFR de Math\'ematiques, Case 7012, 75205 Paris Cedex 13, France}
\email{apredoaei@imj-prg.fr}

\author{\firstname{Xiaonan} \lastname{Ma}}
\address{Chern Institute of Mathematics and LPMC, Nankai University, Tianjin 300071, P.~R.~China}
\email{xiaonan.ma@nankai.edu.cn}

\author{\firstname{Lei} \lastname{Wang}}
\address{School of Mathematics and Statistics, Huazhong University of Science and Technology, Wuhan, Hubei 430074, P.~R.~China}
\email{wanglei2017@hust.edu.cn}

\thanks{X.~M. is partially supported by Nankai Zhide Foundation, ANR-21-CE40-0016 and funded through the Institutional Strategy of the University of Cologne within the German Excellence Initiative. L.~W. is partially supported by NSFC No.~11801187}
\CDRGrant[Nankai Zhide Foundation]{ANR-21-CE40-0016}
\CDRGrant[NSFC]{11801187}

\begin{abstract}
In this paper we consider a punctured Riemann surface endowed with a Hermitian metric that equals the Poincar\'e metric near the punctures, and a holomorphic line bundle that polarizes the metric. We show that the quotient of the induced Fubini--Study forms by Kodaira maps of high tensor powers of the line bundle and the Poincar\'e form near the singularity grows polynomially uniformly on a neighborhood of the singularity as the tensor power tends to infinity, as an application of the method in~\cite{AMM22}.
\end{abstract}

\begin{altabstract}
Dans cet article, nous considÃ©rons une surface de Riemann Ã©pointÃ©e munie d'une mÃ©trique hermitienne qui coÃ¯ncide avec la mÃ©trique de PoincarÃ© prÃ¨s des points de ponction, ainsi qu'un fibrÃ© en droites holomorphe qui polarise la mÃ©trique. Nous montrons que le quotient des formes induites de Fubini--Study par les applications de Kodaira des puissances tensorielles Ã©levÃ©es du fibrÃ© en droites et de la forme de PoincarÃ© prÃ¨s de la singularitÃ© croÃ®t de maniÃ¨re polynomiale et uniforme dans un voisinage de la singularitÃ© lorsque la puissance tensorielle tend vers l'infini, en application de la mÃ©thode dÃ©crite dans~\cite{AMM22}.
\end{altabstract}

\begin{document}

\maketitle

\section{Introduction}

In this paper we study the asymptotics of the induced Fubini--Study metrics by Kodaira maps of high tensor powers of a singular Hermitian line bundle over a Riemann surface under the assumption that the curvature has singularities of Poincar\'e type at a finite set. We show namely that the \emph{quotient} of the induced Fubini--Study metrics by Kodaira maps of high tensor powers of the line bundle and the Poincar\'e model near the singularity grows polynomially uniformly on a neighborhood of the singularity as the tensor power tends to infinity. In~\cite{AMM16,AMM21}, Auvray, Ma and Marinescu obtained a weighted estimate in the $C^m$-norm near the punctures for the difference of the global Bergman kernel and of the Bergman kernel of the Poincar\'e model near the singularity, uniformly in the tensor power $p$ of the given line bundle; and an application in arithmetic geometry is the sharp uniform asymptotics of the sup-norm of automorphic (cusp) forms associated with a Fuchsian group of the first kind which was studied by Ullmo, Kramer... \cite{MiUll98,JK04,fjk}. Moreover, in~\cite{AMM22} they also show that the quotient of the global Bergman kernel and the Bergman kernel of the Poincar\'e model near the singularity tends to one up to $\mathcal{O}(p^{-\infty})$. This paper is an application of the results and the method of~\cite{AMM21,AMM22}.

Following~\cite{Tian}, an important application of the expansion of the Bergman kernel is the convergence of the induced Fubini--Study forms by Kodaira maps. For more works on Bergman kernels, cf.~\cite{Catlin, DLM06, MM08, Zeld, Finski24}, and we refer the readers to the book~\cite{MM07} for a comprehensive study of Bergman kernels. Donaldson~\cite{Donaldson} uses the expansion of the Bergman kernel to study the relation between constant scalar curvature K\"ahler metrics and Chow stability. Coming to our context, an interesting problem is the relation between the existence of special complete/singular metrics and the stability of the pair $(X,D)$ where $D$ is a smooth divisor of a compact K\"ahler manifold $X$; see e.g.\ the suggestions of~\cite[Section~3.1.2]{Szeke} and~\cite{LW}, for the case of ``asymptotically hyperbolic K\"ahler metrics'', which naturally generalize to higher dimensions the complete metrics $\omega$ studied here.

We place ourselves in the setting of~\cite{AMM21} which we describe now. Let $\bar\Sigma$ be a compact Riemann surface and let $D=\{a_1,\dotsc,a_N\}\subset\bar\Sigma$ be a finite set. We consider the punctured Riemann surface $\Sigma = \bar{\Sigma}\smallsetminus D$ and a Hermitian form $\omega_{\Sigma}$ on $\Sigma$. Let $L$ be a holomorphic line bundle on $\bar{\Sigma}$, and let $h$ be a singular Hermitian metric on $L$ such that:
\begin{enumerate}
\renewcommand{\theenumi}{$\alpha$}\relabel
\item \label{alpha} $h$ is smooth over $\Sigma$, and for all $j=1,\dotsc,N$, there is a trivialization of $L$ in the complex neighborhood $\bar{V_j}$ of $a_j$ in $\bar{\Sigma}$, with associated coordinate $z_j$ such that $\abs{1}_{h}^2(z_{j})= \abs[\big]{\log\parens[\big]{\abs{z_j}^2}}$;
\renewcommand{\theenumi}{$\beta$}\relabel
\item \label{beta} there exists $\varepsilon>0$ such that the (smooth) curvature $R^L$ of $h$ satisfies $iR^L\geq\varepsilon\omega_{\Sigma}$ over $\Sigma$ and moreover, $iR^L=\omega_{\Sigma}$ on $V_j \coloneqq \bar{V_j}\smallsetminus\{a_j\}$; in particular, $\omega_{\Sigma} = \omega_{\D^*}$ in the local coordinate $z_j$ on $V_j$ and $(\Sigma, \omega_{\Sigma})$ is complete.
\end{enumerate}
Here $\omega_{\D^*}$ denotes the Poincar\'{e} metric on the punctured unit disc $\D^* = \D \smallsetminus\{0\}$, where $\D$ is the unit disc, normalized as follows:
\begin{equation}\label{eq1}
\omega_{\D^*} \coloneqq \frac{i\,\dd z\wedge \dd \bar{z}} {\abs{z}^2\log^2\parens[\big]{\abs{z}^2}}.
\end{equation}
For $p\geq1$, let $h^p \coloneqq h^{\otimes p}$ be the metric induced by $h$ on $\restr{L^p}{\Sigma}$, where $L^p \coloneqq L^{\otimes p}$. We denote by $H^0_{(2)}(\Sigma,L^p)$ the space of ${\bL}^2$-holomorphic sections of $L^p$ relative to the metrics $h^p$ and $\omega_\Sigma$ endowed with the obvious inner product. By~\cite[(6.2.17)]{MM07}, the sections from $H^0_{(2)}(\Sigma, L^p)$ extend to holomorphic sections of $L^p$ over $\bar\Sigma$ which vanish on $D$. In particular, the dimension $d_p$ of $H^0_{(2)}(\Sigma, L^p)$ is finite.

We denote by $B_p(\,\cdot\,)$ the Bergman kernel function of the orthogonal projection $B_{p}$ from the space of $\bL^{2}$-sections of $L^{p}$ over $\Sigma$ onto $H^0_{(2)}(\Sigma,L^p)$. If $\{S_\ell^p\}_{\ell=1}^{d_p}$ is an orthonormal basis of $H^0_{(2)}(\Sigma,L^p)$, then
\begin{equation}\label{eq4}
B_p(x) \coloneqq \sum_{\ell=1}^{d_p}\abs[\big]{S^p_\ell(x)}_{h^p}^2\,.
\end{equation}
Note that these are independent of the choice of basis (see~\cite[(6.1.10)]{MM07}). Similarly, let $B_p^{\D^*}(x)$ be the Bergman kernel function of $\parens[\big]{\D^*, \omega_{\D^*},\C,\abs[\big]{\log\parens[\big]{\abs{z}^2}}^p\, h_{0}}$ with $h_{0}$ the flat Hermitian metric on the trivial line bundle $\C$.

We fix a point $\mathbf{a}\in D$ and work in coordinates centered at $\mathbf{a}$. Let $\mathfrak{e}_L$ be the holomorphic frame of~$L$ near $\mathbf{a}$ corresponding to the trivialization in the condition~\eqref{alpha}. By the assumptions~\eqref{alpha} and~\eqref{beta} we have the following identification of the geometric data in the coordinate $z$ on the punctured disc $\mathbb{D}^*_{4r}$ of radius $4r$ centered at $\mathbf{a}$, via the trivialization $\mathfrak{e}_L$ of $L$,
\begin{equation}\label{eq8}
\restr{\parens[\big]{\Sigma, \omega_{\Sigma}, L,h}}{\mathbb{D}^*_{4r}} = \restr{\parens[\Big]{\mathbb{D}^*, \omega_{\mathbb{D}^*}, \mathbb{C}, h_{\mathbb{D}^*} = \abs[\big]{\log\parens[\big]{\abs{z}^2}}
\cdot h_0}}{\mathbb{D}^*_{4r}}, \quad
\text{with $0< r < (4e)^{-1}$}.
\end{equation}
In~\cite{AMM22}, Auvray, Ma and Marinescu proved the following estimates.

\begin{theo}[{\cite[Theorems~1.2 and~1.3]{AMM22}}]\label{thm_prevThm}
If\/ $(\Sigma, \omega_\Sigma, L, h)$ fulfill conditions~\eqref{alpha} and~\eqref{beta}, then for all $k\in \mathbb{N}$, $l\in \mathbb{N}^*$ and $D_1,\dotsc, D_k \in \braces[\big]{ \frac{\partial}{\partial z},
\frac{\partial}{\partial \bar{z}}}$, there exists $C>0$ such that for any $p\in\N^{*}$ we have
\begin{equation}\label{eq1.7}
\sup_{z\in V_1\cup\dotsb\cup V_N}
\abs[\Bigg]{(D_1 \dotsm D_k) \parens[\Bigg]{\frac{B_p}{B_p^{\D^*}}(z) - 1}} \leq C p^{-\ell}.
\end{equation}
\end{theo}

Let us consider the Kodaira map at level $p\geq 2$ induced by $H^0_{(2)}(\Sigma, L^p)$, which is a meromorphic map defined by
\begin{equation}\label{eq5}
J_{p,(2)} \colon \Sigma \dashrightarrow
\mathbb{P}\parens[\big]{H^0_{(2)}(\Sigma, L^p)^*}
\cong \mathbb{CP}^{d_p-1}, \quad x\mapsto
\braces[\big]{ \sigma \in H^0_{(2)}(\Sigma, L^p): \sigma(x) = 0}.
\end{equation}
By~\cite[p.~2364]{AMM22}, $J_{p,(2)}$ is an embedding for $p\gg 1$.

The ${\bL}^2$-metric on $H^0_{(2)}(\Sigma, L^p)$ induces a Fubini--Study form $\omega_{\FS, p}$ on the projective space $\mathbb{P}\parens[\big]{H^0_{(2)}(\Sigma, L^p)^*}$. We have by~\cite[Theorem~5.1.3, (5.1.21)]{MM07}
\begin{equation}\label{eq6}
\frac{1}{p} J^*_{p,(2)} \omega_{\FS,p} = \frac{i}{2\pi} R^L + \frac{i}{2\pi p} \partial \bar{\partial} \log (B_p).
\end{equation}
From~\eqref{eq1.7} and~\eqref{eq6} (cf.~\cite[Theorem~4.1]{AMM22}), we have, as $p\rightarrow +\infty$,
\begin{equation}\label{eq7}
\frac{1}{p} J^*_{p,(2)} \omega_{\FS,p} = \frac{1}{2\pi} \omega_{\Sigma} + \frac{i}{2\pi p} \partial \bar{\partial}
\log (B_p^{\mathbb{D}^* }) + \mathcal{O}(p^{-\infty}),
\end{equation}
uniformly on $V_1 \cup V_2 \cup \dotsb \cup V_N$. Note that, by condition~\eqref{beta}, we have
\begin{equation}\label{eq001}
\omega_{\D^*} = \omega_{\Sigma} = i R^L = - i\,\partial\bar{\partial} \log \abs[\big]{\log\parens[\big]{\abs{z}^2}} \quad
\text{on $V_j$}.
\end{equation}

The main result of the present paper is the following estimate of the quotient of the induced Fubini--Study forms $\frac{1}{p} J^*_{p,(2)} \omega_{\FS,p}$ by Kodaira maps of $L^p$ and the Poincar\'e form $\omega_{\mathbb{D}^*}(z)$.
\begin{theo}\label{thm_MainThm}
If\/ $(\Sigma, \omega_{\Sigma}, L, h)$ fulfill conditions~\eqref{alpha} and~\eqref{beta}, then
\begin{equation}\label{eq9}
\sup_{z\in V_1 \cup \dotsb \cup V_N }
\abs*{\frac{ J^*_{p,(2)} \omega_{\FS,p}(z) } {p \, \omega_{\mathbb{D}^*} (z) }} = \mathcal{O}(p^3) \quad \text{as $p\rightarrow +\infty$}.
\end{equation}
\end{theo}

Let us mention that the difficulty of the estimate~\eqref{eq9} consists in the fact that $B_p^{\mathbb{D}^*}(\,\cdot\,)$ vanishes at~$0$. Moreover, by~\cite[Theorem~6.1.1]{MM07} and~\eqref{eq6}, for any $K \subset \Sigma$ compact,
\begin{equation}\label{eq0006}
\frac{1}{p} J^*_{p,(2)} \omega_{\FS,p} - \frac{i}{2\pi} R^L = \mathcal{O}(p^{-1}) \quad \text{on $K$}.
\end{equation}

We give an important example where Theorem~\ref{thm_MainThm} applies. Let $\Gamma \subset \PSL(2, \mathbb{R})$ be a geometrically finite Fuchsian group of the first kind without elliptic elements. Then $\Sigma\coloneqq \Gamma \backslash \mathbb{H}$ can be compactified by finitely many points $D= \{a_1,\dotsc, a_N\}$ into a compact Riemann surface $\bar{\Sigma}$~\cite[p.~956]{AMM21}. Let $\omega_{\Sigma}$ be the KÃ¤hler--Einstein metric on $\Sigma$ induced by the Poincar\'e metric $\omega_{\mathbb{H}} = \frac{ i\,\dd z \wedge \dd \bar{z}}{ 4\abs{\Imag z}^2}$ on the upper-half plane $\mathbb{H}$.

Let $\mathcal{S}^{\Gamma}_{2p}$ be the space of cusp forms (Spitzenformen) of weight $2p$ of $\Gamma$ endowed with the Petersson inner product (cf.~\cite[p.~995]{AMM21}). We can form the Bergman kernel function of $\mathcal{S}_{2p}^{\Gamma}$ as in~\eqref{eq4}, denoted by $S_{p}^{\Gamma}$ \cite[p.~995]{AMM21}, then
\begin{equation}\label{eq10a}
S_{p}^{\Gamma}(z) = \sum_j \abs[\big]{f_j(z)}^2 (2 \Imag z)^{2p}
\end{equation}
with $\{f_j\}$ any orthonormal basis of $\mathcal{S}^{\Gamma}_{2p}$. For any $p>0$ and $z\in \Sigma$, define
\begin{equation}\label{eq.berg}
\omega_{\Sigma}^{\Ber, p} \coloneqq 
\frac{i}{2\pi} \partial \bar{\partial}
\log \parens[\big]{S_{p}^{\Gamma}(z)}.
\end{equation}
As a corollary of Theorem~\ref{thm_MainThm}, we get the following:
\begin{coro}\label{cor}
Let $\Gamma \subset \PSL(2, \mathbb{R})$ be a geometrically finite Fuchsian group of the first kind without elliptic elements and\/ $\Sigma \coloneqq \Gamma \backslash \mathbb{H}$. Then
\begin{equation}\label{eq10}
\sup_{z\in \Sigma} \abs*{\frac{ \omega_{\Sigma}^{\Ber, p}(z) } { p \, \omega_{\Sigma}(z)}} = \mathcal{O}(p^3) \quad \text{as $p\rightarrow +\infty$}.
\end{equation}
\end{coro}

\begin{rema}
Corollary~\ref{cor} still holds if $\Gamma$ has elliptic elements. In this case, the quotient $\Sigma = \Gamma \backslash \mathbb{H}$ is an orbifold, by the result of Dai--Liu--Ma~\cite[(5.25)]{DLM06}, Ma--Marinescu~\cite[Section~5.4.3]{MM07} on the Bergman kernel on orbifolds and the localization of the asymptotics of the Bergman kernel on compact part of $\Sigma$, from~\cite[(1.15), p.~998]{AMM21}, we know near the orbifold points, $\mathcal{O}(p^3)$ in~\eqref{eq10} can be estimated as $\mathcal{O}(1)$.
\end{rema}

\begin{rema}
In~\cite[Main Theorem~1]{Arya23}, the authors claim a stronger estimate than~\eqref{eq10} and they prove their estimate by using a computation on classical modular forms, but unfortunately their proof is not complete. As a comparison, the difficulty in proving~\eqref{eq10} arises from the fact that the Bergman kernel function vanishes on $D$, but~\cite[Proposition~2.3]{Arya23} claims that it has a uniform strictly positive lower bound, cf.~\cite[Corollary~3.6]{AMM21} and the explanation after~\eqref{eq16}.

In~\cite[Main Theorem~2]{Arya24}, the authors claim a high dimension version of~\eqref{eq10} for ball quotients, but unfortunately their proof is also not complete. The similar incomplete argument occurs in the proof of~\cite[Proposition~3.6]{Arya24}.
\end{rema}

\section{Proof of Theorem~\ref{thm_MainThm} and Corollary~\ref{cor}}

In this section, we establish Theorem~\ref{thm_MainThm} and Corollary~\ref{cor} by using the techniques in~\cite{AMM22}. Our starting point is~\eqref{eq7}, and thus we only need to work on the punctured unit disc $\mathbb{D}^*$. The key part in our proof of Theorem~\ref{thm_MainThm} in Section~\ref{s2.1} is Lemma~\ref{lem} which is established in Section~\ref{s2.2}: for the four terms in~\eqref{eq26} and~\eqref{eq27}, we get $\mathcal{O}(p^{-\infty})$. The main estimate $\mathcal{O}(p^3)$ is from~\eqref{eq28} as $B_p^{\D^*}(z)$ oscillates too violently near $\abs{z}=e^{-p}$ (cf.~\cite[Section~3.2]{AMM21}). In Section~\ref{s2.3}, we explain the proof of Corollary~\ref{cor}.

\subsection{Proof of Theorem~\ref{thm_MainThm}} \label{s2.1}

From~\cite[(2.6) and~(2.7)]{AMM22}, we have
\begin{equation}\label{eq15}
B_p^{\mathbb{D}^*}(z) = \abs[\big]{\log\parens[\big]{\abs{z}^2}}^p\beta_p^{\mathbb{D}^*}(z),
\quad \text{for $z\in \mathbb{D}^*$},
\end{equation}
where
\begin{equation}\label{eq16}
\beta_p^{\mathbb{D}^*}(z) = \sum_{l=1}^{\infty} (c_l^{(p)})^2 \abs{z}^{2l}, \quad
\text{with $c_l^{(p)} = \parens[\Bigg]{\frac{l^{p-1}}{2\pi(p-2)\mathrel{!}}}^{1/2}$}.
\end{equation}
From~\eqref{eq15} and~\eqref{eq16}, we know $\lim_{z\rightarrow 0} B_p^{\D^*}(z) =0$ for any $p$. Thus, $B_p^{\D^*}$ has no strictly positive lower bound. This is the main difficulty to establish Theorem~\ref{thm_prevThm}.

By~\cite[Proposition~3.3]{AMM21}, for any $m\in \mathbb{N}$, $0<b <1$ and $0<\gamma <\frac{1}{2}$, there exists $\varepsilon = \varepsilon(b,\gamma) >0$ such that
\begin{equation}\label{eq12}
\norm[\Bigg]{B_p^{\mathbb{D}^*} (z) - \frac{p-1}{2\pi}}_{ C^m(\{be^{- p^{\gamma}}
\leq \abs{z} <1\}, \omega_{\mathbb{D}^*})} = \mathcal{O}(e^{-\varepsilon p^{1-2\gamma}}) \quad \text{as $p\rightarrow +\infty$}.
\end{equation}

After reducing to some $V_j$ and identifying the geometric data on $\mathbb{D}^*_{4r}$ and $\Sigma$ via~\eqref{eq8}, by~\eqref{eq1} and~\eqref{eq7}, we have
\begin{equation}\label{eq11}
\begin{split}
\frac{ J^*_{p,(2)} \omega_{\FS,p}(z) } { p \,
\omega_{\mathbb{D}^*} (z) } \!-\! \frac{1}{2\pi} \!
& = \! \frac{ i \,\partial\bar{\partial}
\log \parens[\big]{B^{\mathbb{D}^*}_p(z)}}{ 2\pi p \,
\omega_{\mathbb{D}^*}(z) } + \mathcal{O}(p^{-\infty}) \\
& = \! \frac{1}{2\pi p} \abs{z}^2 \parens[\big]{\log\parens[\big]{\abs{z}^2}}^2
\frac{ B_p^{\mathbb{D}^*} \!\!\! (z) \cdot
\frac{\partial^2}{\partial z \,\partial\bar{z}} B^{\mathbb{D}^*}_p \!\!\! (z) - \frac{\partial}{\partial z} B_p^{\mathbb{D}^*} \!\!\! (z) \cdot \frac{\partial}{\partial \bar{z}} B^{\mathbb{D}^*}_p \!\!\! (z) }{\parens[\big]{B_p^{\mathbb{D}^*}(z)}^2} + \mathcal{O}(p^{-\infty}).
\end{split}
\end{equation}
As $\abs{z}\log\abs{z}^2 \frac{\partial}{\partial z}$ is an orthonormal frame of $(T^{(1,0)} \D^*, \omega_{\D^*})$, from~\eqref{eq12} with $m=1,2$, we have
\begin{equation}\label{eq003}
\begin{aligned}
& \sup_{ be^{- p^{\gamma}} \leq \abs{z} <1 } \abs{z}^2
\parens[\big]{\log\parens[\big]{\abs{z}^2}}^2 \abs[\Bigg]{\frac{\partial^2} {\partial z \,\partial\bar{z}} B^{\mathbb{D}^*}_p(z)} = \mathcal{O}(e^{-\varepsilon p^{1-2\gamma}}) \quad \text{as $p\rightarrow +\infty$}, \\
& \sup_{ be^{- p^{\gamma}} \leq \abs{z} <1 } \abs{z}
\abs[\big]{\log\parens[\big]{\abs{z}^2}} \abs[\Bigg]{\frac{\partial}{\partial z} B_p^{\mathbb{D}^*}(z)} = \mathcal{O}(e^{-\varepsilon p^{1-2\gamma}}) \quad \text{as $p\rightarrow +\infty$}.
\end{aligned}
\end{equation}
From~\eqref{eq12} with $m=0$, \eqref{eq11} and~\eqref{eq003}, we get
\begin{equation}\label{eq13}
\sup_{ be^{- p^{\gamma}} \leq \abs{z} <1 } \abs*{\frac{ J^*_{p,(2)} \omega_{\FS,p}(z) } { p\, \omega_{\mathbb{D}^*} (z) } - \frac{1}{2\pi}} = \mathcal{O}(p^{-\infty}).
\end{equation}
Thus, in order to prove Theorem~\ref{thm_MainThm} it suffices to show that
\begin{equation}\label{eq14}
\sup_{ 0 < \abs{z} < be^{- p^{\gamma}}}
\abs*{\frac{ J^*_{p,(2)} \omega_{\FS,p}(z) } { p\, \omega_{\mathbb{D}^*} (z)}} = \mathcal{O}(p^3).
\end{equation}
From~\eqref{eq1}, \eqref{eq7}, \eqref{eq001} and~\eqref{eq15}, we get
\begin{equation}\label{eq19}
\frac{ J^*_{p,(2)} \omega_{\FS,p} (z) }{ p\,
\omega_{\mathbb{D}^*}(z) } = \abs{z}^2 \parens[\big]{\log \abs{z}^2}^2 \cdot
\frac{ \frac{\partial^2}{\partial z\,\partial\bar{z}}
\beta_p^{\mathbb{D}^*}(z) \cdot \beta_p^{\mathbb{D}^*} (z) - \frac{\partial}{\partial z} \beta_p^{\mathbb{D}^*}(z)
\frac{\partial}{\partial \bar{z}} \beta_p^{\mathbb{D}^*}(z) } {2\pi p \parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 } + \mathcal{O}(p^{-\infty}).
\end{equation}

A key idea of~\cite{AMM22} is that we use the precise formula~\eqref{eq16}, and only the lower degree (i.e., $l=\mathcal{O}(p)$) part of~\eqref{eq16} plays the essential contribution in~\eqref{eq19}. We define $\delta_p$ as follows,
\begin{equation}\label{eq22}
\delta_p = \Bigg\lfloor \frac{(p-2) }{ 2 \abs{\log r}} \Bigg\rfloor,
\end{equation}
and $\lfloor x\rfloor$ is the integral part of $x\in \mathbb{R}$. Set
\begin{equation}\label{eq21}
\begin{aligned}
& I_{1,p} \coloneqq \sum_{l=1}^{\delta_p} \sum_{m=1}^{\delta_p} l(l-m) (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m}, \\
& I_{2,p} \coloneqq \sum_{l=1}^{\delta_p} \sum_{m=\delta_p +1}^{\infty} l(l-m) (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m}, \\
& I_{3,p} \coloneqq \sum_{l=\delta_p +1}^{\infty} \sum_{m=1}^{\delta_p} l(l-m) (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m}, \\
& I_{4,p} \coloneqq \sum_{l=\delta_p+1}^{\infty}
\sum_{m=\delta_p+1}^{\infty} l(l-m) (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m}.
\end{aligned}
\end{equation}
From~\eqref{eq16} and~\eqref{eq21}, we have
\begin{equation}\label{eq20}
\begin{split}
& \abs{z}^2 \parens*{\frac{\partial^2}{\partial z\,\partial\bar{z}}
\beta_p^{\mathbb{D}^*}(z) \cdot \beta_p^{\mathbb{D}^*} (z) - \frac{\partial}{\partial z} \beta_p^{\mathbb{D}^*}(z)\frac{\partial}{\partial \bar{z}} \beta_p^{\mathbb{D}^*}(z)}\\
& \hspace{2cm} = \abs{z}^2 \sum_{l,m=1}^{\infty} l^2 (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m-2} - \abs{z}^2 \sum_{l,m=1}^{\infty} lm(c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m-2} \\
& \hspace{2cm} = \sum_{l,m=1}^{\infty} l(l-m) (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m} \\
& \hspace{2cm} = I_{1,p} + I_{2,p} + I_{3,p} + I_{4,p}.
\end{split}
\end{equation}
Thus, by~\eqref{eq19} and~\eqref{eq20} we get
\begin{equation}\label{eq24}
\frac{ J^*_{p,(2)} \omega_{\FS,p} (z) }{ p \,
\omega_{\mathbb{D}^*}(z) } = \parens[\big]{\log \abs{z}^2}^2 \cdot \frac{ I_{1,p} + I_{2,p} + I_{3,p} + I_{4,p}} { 2\pi p
\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2} +\mathcal{O}(p^{-\infty}).
\end{equation}
We will show the following result.
\begin{lemm}\label{lem}
\mbox{}
\begin{enumerate}
\item \label{lem_1} For $j=2,3,4$, we have
\begin{equation}\label{eq26}
\sup_{ 0 < \abs{z} < be^{- p^{\gamma}}}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{j,p}}{ \parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }} = \mathcal{O}(p^{-\infty}).
\end{equation}
\item \label{lem_2} We have
\begin{subequations}
\begin{equation}\label{eq27}
\sup_{ 0 < \abs{z} < 2 e^{-p}}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{1,p}}{ \parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }} = \mathcal{O}(p^{-\infty}),
\end{equation}
\begin{equation}\label{eq28}
\sup_{ e^{-p} < \abs{z} < be^{- p^{\gamma}}}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{1,p}}{ \parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }} = \mathcal{O}(p^4).
\end{equation}
\end{subequations}
\end{enumerate}
\end{lemm}

\begin{enonce*}[remark]{How to get Theorem~\ref{thm_MainThm} from Lemma~\ref{lem}}
Since $\braces[\big]{z\in \mathbb{C}: 0<\abs{z} < be^{-p^{\gamma}}} = \braces[\big]{z \in \mathbb{C}: 0<\abs{z} < 2e^{-p}} \cup \braces[\big]{z\in \mathbb{C}: e^{-p} < \abs{z} < be^{-p^{\gamma}}}$, from~\eqref{eq27} and~\eqref{eq28}, we get
\begin{equation}\label{eq006}
\sup_{ 0 < \abs{z} < be^{- p^{\gamma}}}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{1,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }} = \mathcal{O}(p^4).
\end{equation}
From~\eqref{eq24}, \eqref{eq26} and~\eqref{eq006}, we get~\eqref{eq14}. The proof of Theorem~\ref{thm_MainThm} is completed.
\end{enonce*}

\subsection{Proof of Lemma~\ref{lem}}\label{s2.2}

From~\eqref{eq21}, we have
\begin{equation}\label{eq29}
\abs{I_{1,p}} \leq \sum_{m=2}^{\delta_p} \abs{1-m} (c_1^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2+2m} + \sum_{l=2}^{\delta_p} \sum_{m=2}^{\delta_p } l\abs{l-m} (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m}.
\end{equation}

We first estimate the first term on the right hand side of~\eqref{eq29}. By~\eqref{eq16}, we have
\begin{equation}\label{eq30}
(c^{(p)}_1)^2 \abs{z}^2 < \beta_p^{\mathbb{D}^*}(z),
\end{equation}
and
\begin{equation}\label{eq31}
(c^{(p)}_m)^2 = \frac{ m^{p-1}}{ 2\pi(p-2)\mathrel{!} } = \parens[\Bigg]{\frac{m}{m-1}}^{p-1} \cdot
\frac{ (m-1)^{p-1}}{ 2\pi(p-2)\mathrel{!} } = \parens[\Bigg]{\frac{m}{m-1}}^{p-1} (c^{(p)}_{m-1})^2.
\end{equation}
From~\eqref{eq16} and~\eqref{eq31}, as $\frac{l}{l-1} \leq 2$ for $l\geq 2$, we have
\begin{equation}\label{eq0038}
\begin{split}
\sum_{l=2}^{\delta_p} (c_l^{(p)})^2 \abs{z}^{2(l-1)}
& = \sum_{l=2}^{\delta_p} \parens[\Bigg]{\frac{l}{l-1}}^{p-1}
(c_{l-1}^{(p)})^2 \abs{z}^{2(l-1)} \\
& \leq 2^{p-1} \sum_{l=2}^{\delta_p}
(c_{l-1}^{(p)})^2 \abs{z}^{2(l-1)}
\leq 2^{p-1} \beta_p^{\mathbb{D}^*} (z).
\end{split}
\end{equation}
From~\eqref{eq22}, \eqref{eq30} and~\eqref{eq0038}, we have
\begin{equation}\label{eq38}
\begin{split}
\sum_{m=2}^{\delta_p} \abs{1-m} (c_1^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2+2m}
& \leq (\delta_p-1) (c_1^{(p)})^2 \abs{z}^2 \sum_{m=2}^{\delta_p} (c_m^{(p)})^2 \abs{z}^{2(m-1)} \cdot \abs{z}^2 \\
& \leq (\delta_p -1) 2^{p-1} \abs{z}^2 \, \beta_p^{\mathbb{D}^*}(z)^2 \\
& \leq p \, 2^{p-1} \abs{z}^2 \beta_p^{\mathbb{D}^*}(z)^2.
\end{split}
\end{equation}

\begin{proof}[Proof of~(\ref{eq27})]
For $0<\abs{z}<2 e^{-p}$, we have
\begin{equation}\label{eq35}
2^{p-1} \abs{z} < 2^p e^{-p} = \parens[\Bigg]{\frac{2}{e}}^p.
\end{equation}
Thus, by~\eqref{eq38} and~\eqref{eq35}, for $\abs{z} < 2 e^{-p}$, we get
\begin{equation}\label{eq36}
\sum_{m=2}^{\delta_p} \abs{1-m} (c_1^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2+2m}
\leq p \parens[\Bigg]{\frac{2}{e}}^p \abs{z}
\,\beta_p^{\mathbb{D}^*}(z)^2.
\end{equation}
Next we estimate the second term on the right hand side of~\eqref{eq29}:
\begin{equation}\label{eq37}
\begin{split}
\sum_{l=2}^{\delta_p} \sum_{m=2}^{\delta_p } l\abs{l-m} (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m}
& \leq \delta_p \cdot \delta_p \sum_{l=2}^{\delta_p}
\sum_{m=2}^{\delta_p} (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m}
\quad \text{(as $\abs{l-m}<\delta_p$)}\\
& = \delta_p^2 \, \abs{z}^4 \sum_{l=2}^{\delta_p} (c_l^{(p)})^2 \abs{z}^{2(l-1)} \cdot \sum_{m=2}^{\delta_p} (c_m^{(p)})^2 \abs{z}^{2(m-1)}.
\end{split}
\end{equation}
By~\eqref{eq22}, \eqref{eq0038}, \eqref{eq35} and~\eqref{eq37}, for $\abs{z}<2e^{-p}$, as in~\eqref{eq36}, we get
\begin{equation}\label{eq39}
\begin{split}
\sum_{l=2}^{\delta_p} \sum_{m=2}^{\delta_p } l\abs{l-m} (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m}
& \leq \delta_p^2 \, \abs{z}^4 \cdot 2^{p-1} \beta_p^{\mathbb{D}^*} (z) \cdot 2^{p-1} \beta_p^{\mathbb{D}^*} (z) \\
& \leq p^2 \cdot \parens[\Bigg]{\frac{2}{e}}^{2p} \abs{z}^2 \,
\beta_p^{\mathbb{D}^*}(z)^2.
\end{split}
\end{equation}
Combining~\eqref{eq29}, \eqref{eq36} and~\eqref{eq39}, we obtain
\begin{equation}\label{eq41}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{1,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }}
\leq p \parens[\Bigg]{\frac{2}{e}}^p \abs{z} \parens[\big]{\log \abs{z}^2}^2 + p^2 \parens[\Bigg]{\frac{2}{e}}^{2p} \abs{z}^2
\parens[\big]{\log \abs{z}^2}^2.
\end{equation}
Since the functions $f(x) = x(\log x^2)^2$ and $g(x) = x^2(\log x^2)^2$ are bounded on the interval $\mathopen{]}0, 2 e^{-p}\mathclose{[}\subset\mathopen{]}0,e^{-2}\mathclose{[}$, by~\eqref{eq41}, we get that there exists $C_1 >0$ such that
\begin{equation}\label{eq42}
\sup_{ 0<\abs{z}< 2 e^{-p}}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{1,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }}
\leq C_1 \bracks*{p\parens[\Bigg]{\frac{2}{e}}^p + p^2 \parens[\Bigg]{\frac{2}{e}}^{2p}}.
\end{equation}
By~\eqref{eq42}, we get~\eqref{eq27}.
\let\qed\relax
\end{proof}

\begin{proof}[Proof of~(\ref{eq28})]
From~\eqref{eq16}, \eqref{eq22} and~\eqref{eq21}, we have
\begin{equation}\label{eq43}
\begin{split}
\abs{I_{1,p}}
& \leq \sum_{l=1}^{\delta_p} \sum_{m=1}^{\delta_p } l\abs{l-m} (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m} \\
& \leq \delta_p \cdot \delta_p \sum_{l=1}^{\delta_p} \sum_{m=1}^{\delta_p } (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m} \\
& \leq p^2 \beta_p^{\mathbb{D}^*}(z)^2.
\end{split}
\end{equation}
Therefore,
\begin{equation}\label{eq44}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{1,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }}
\leq p^2 \parens[\big]{\log\abs{z}^2}^2.
\end{equation}
For $z \in \braces[\big]{u \in \mathbb{C} : e^{-p} <\abs{u}< b e^{-p^{\gamma}}}$, we have $e^{-2p} < \abs{z}^2 <1$. Thus, $-2p < \log \abs{z}^2 <0$. Therefore, $\parens[\big]{\log \abs{z}^2}^2 < 4p^2$. Hence, by~\eqref{eq44}, we have
\begin{equation}\label{eq45}
\sup_{ e^{-p} < \abs{z} < b e^{-p^{\gamma}}}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{1,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }}
\leq p^2 \cdot 4p^2 = 4 p^4.
\end{equation}
From~\eqref{eq45}, we get~\eqref{eq28}.
\let\qed\relax
\end{proof}

\begin{proof}[Proof of Lemma~\ref{lem}(\ref{lem_1}) for $\boldsymbol{j=2}$]
From~\eqref{eq21}, we have
\begin{equation}\label{eq46}
\begin{split}
\abs{I_{2,p}}
& \leq \sum_{l=1}^{\delta_p} \sum_{m=\delta_p +1}^{\infty} l\abs{l-m} (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m} \\
& < \sum_{l=1}^{\delta_p} l (c_l^{(p)})^2 \abs{z}^{2l} \cdot
\sum_{m=\delta_p +1}^{\infty} m (c_m^{(p)})^2 \abs{z}^{2m},
\end{split}
\end{equation}
as $l\abs{l-m} =l(m-l) < lm$.

From~\cite[(3.70)]{AMM22}, there exists $C>0$ such that for any $s\in\mathbb{N}^*$, $p\geq 2$, we have
\begin{equation}\label{eq48}
\abs{c_s^{(p)}} \leq C p^{\frac{1}{2}} \frac{1}{(2r)^s}
\frac{1}{ \parens[\big]{\log \abs{2r}^2}^{p/2}}.
\end{equation}

Next, we need the following.
\begin{lemm}[{\cite[(3.60)]{AMM22}}]\label{lem2}
For any $\tau \in \mathbb{N}$ fixed, we have
\begin{equation}\label{eq52}
\frac{1}{ \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^{p/2}}
\parens[\Bigg]{\frac{\abs{z}}{ 2r }}^{\delta_p -\tau +1}
\leq C p^{-\frac{1}{2}} \frac{1}{2^{\alpha' p}}
\beta_p^{\mathbb{D}^*}(z)^{\frac{1}{2}}
\end{equation}
for all $p\gg 1$ and $\abs{z} \leq c' p^{-A'}$, where $c'$ and $A'$ are constants defined as follows:
\begin{equation}\label{eq-lem1}
A' =\frac{1}{2\alpha'}, \quad \alpha' =\frac{1}{4\abs{\log r}} \quad \text{and} \quad c'= r e^{1/2\alpha'} \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^{1/2\alpha'}.
\end{equation}
\end{lemm}

\begin{proof}
For the sake of completeness, we include
a proof here. By~\eqref{eq22}, for any $\tau \in \mathbb{N}$ fixed, we obtain
\begin{equation}\label{eq-lem2}
\alpha' p \leq \delta_p - \tau \quad \text{for $p \gg 1$}.
\end{equation}
Thus, by~\eqref{eq-lem1} and~\eqref{eq-lem2} for $\tau \in \mathbb{N}$ fixed, we have
\begin{equation}\label{eq-lem3}
\parens[\Bigg]{\frac{\abs{z}}{2r}}^{2(\delta_p -\tau)/p}
\frac{1} {\abs[\big]{\log\parens[\big]{\abs{2r}^2}}}
\leq \parens[\Bigg]{\frac{\abs{z}}{2r}}^{2\alpha'}
\frac{1} {\abs[\big]{\log\parens[\big]{\abs{2r}^2}}}
\leq 2^{-2\alpha'} \frac{e}{p}
\end{equation}
for $p\gg 1$, $\abs{z} \leq c' p^{-A'}$. Recall that the Stirling formula states
\begin{equation}\label{eq-lem4}
\frac{p^p}{p\mathrel{!}} = (2\pi p)^{-1/2} e^p
\parens[\big]{1 + \mathcal{O}(p^{-1})} \quad
\text{as $p\rightarrow + \infty$}.
\end{equation}
By~\eqref{eq16}, \eqref{eq-lem3} and~\eqref{eq-lem4}, for any $\tau \in \mathbb{N}$ fixed, we have
\begin{equation}
\begin{split}
\frac{1}{ \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^{p/2}} \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^{\delta_p -\tau +1}
& = \frac{1}{2r} \parens*{\parens[\Bigg]{\frac{\abs{z}}{2r}}^{2(\delta_p -\tau)/p} \frac{1} {\abs[\big]{\log\parens[\big]{\abs{2r}^2}}}}^{p/2} \parens[\big]{2\pi (p-2)\mathrel{!}}^{1/2} c_1^{(p)}\abs{z} \\
& \leq C p^{-\frac{1}{2}} \frac{1}{2^{\alpha' p}}
\beta_p^{\mathbb{D}^*}(z)^{\frac{1}{2}}
\end{split}
\end{equation}
for $p\gg 1$, $\abs{z} \leq c' p^{-A'}$.
\end{proof}

From now on, we assume $\abs{z}\leq c' p^{-A'}$, $p \gg 1$. By~\eqref{eq16} and~\eqref{eq22} we get
\begin{equation}\label{eq55}
\sum_{l=1}^{\delta_p} l (c_l^{(p)})^2 \abs{z}^{2l}
\leq \delta_p \sum_{l=1}^{\delta_p} (c_l^{(p)})^2 \abs{z}^{2l}
\leq \delta_p \beta_p^{\mathbb{D}^*}(z)
\leq p \beta_p^{\mathbb{D}^*}(z).
\end{equation}
On the other hand, by~\eqref{eq48}, we have
\begin{equation}\label{eq56}
\begin{split}
\sum_{m=\delta_p+1}^{\infty} m (c_m^{(p)})^2 \abs{z}^{2m}
& \leq \sum_{m=\delta_p +1}^{\infty} m C p \frac{1}{(2r)^{2m}}
\frac{1}{ \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^p } \abs{z}^{2m} \\
& = C p \frac{1}{ \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^p }
\sum_{m=\delta_p +1}^{\infty} m \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^{2m}.
\end{split}
\end{equation}
Since for $0 \leq \xi <1$,
\begin{equation*}
\sum_{q=N+1}^{\infty} q \xi^{q-1} = \parens*{\sum_{q=N+1}^{\infty} \xi^q}' 
= \parens[\Bigg]{\frac{ \xi^{N+1}}{ 1 -\xi }}' = \frac{ (N+1) \xi^N - N \xi^{N+1}}{ (1-\xi)^2 }
\leq \frac{ (N+1) \xi^N } { (1-\xi)^2 },
\end{equation*}
we have
\begin{equation}\label{eq57}
\begin{split}
\sum_{m=\delta_p +1}^{\infty} m \parens[\Bigg]{\frac{\abs{z}}{2r}}^{2m}
& = \parens[\Bigg]{\frac{\abs{z}}{2r}}^2 \sum_{m=\delta_p +1}^{\infty} m \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^{2(m-1)} \\
& \leq \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2 (\delta_p + 1)
\frac{ \parens[\Big]{\frac{\abs{z}}{2r}}^{2\delta_p}} {
\parens[\Big]{1 - \parens[\Big]{\frac{\abs{z}}{2r}}^2}^2 } \\
& \leq \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2 (\delta_p + 1)
\cdot \parens[\Bigg]{\frac{4}{3}}^2 \parens[\Bigg]{\frac{\abs{z}}{2r}}^{2\delta_p}.
\end{split}
\end{equation}
Therefore, by~\eqref{eq22}, \eqref{eq52} for $\tau=1$, \eqref{eq56} and~\eqref{eq57}, we obtain
\begin{equation}\label{eq58}
\begin{split}
\sum_{m=\delta_p+1}^{\infty} m (c_m^{(p)})^2 \abs{z}^{2m}
& \leq \parens[\Bigg]{\frac{4}{3}}^2 \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2 (\delta_p + 1)\cdot C p \frac{1}{ \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^p }
\parens[\Bigg]{\frac{\abs{z}}{2r}}^{2\delta_p} \\
& \leq \parens[\Bigg]{\frac{4}{3}}^2 \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2 (p + 1) C' \frac{1}{2^{2\alpha' p}} \beta_p^{\mathbb{D}^*}(z).
\end{split}
\end{equation}
From~\eqref{eq46}, \eqref{eq55} and~\eqref{eq58}, we get
\begin{equation}\label{eq61}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{ I_{2,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }} \leq
\parens[\big]{\log \abs{z}^2}^2 \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2
\parens[\Bigg]{\frac{4}{3}}^2 C' p (p + 1)
\frac{1}{2^{2\alpha' p}}.
\end{equation}
Since the function $f(x) = x^2(\log x^2)^2$ is bounded on the interval $\mathopen{]}0,1\mathclose{[}$, from~\eqref{eq61}, we get
\begin{equation}\label{eq62}
\sup_{ 0<\abs{z} \leq c' p^{-A'}}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{2,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }}
\leq C'' p^2 \frac{1}{2^{2\alpha' p}}.
\end{equation}
Since $c' p^{-A'} > b e^{-p^{\gamma}}$ for $p\gg 1$, we get~\eqref{eq26} for $j=2$ from~\eqref{eq62}.
\let\qed\relax
\end{proof}

\begin{proof}[Proof of Lemma~\ref{lem}(\ref{lem_1}) for $\boldsymbol{j=3}$]
We assume $\abs{z}\leq c' p^{-A'}$, $p \gg 1$ as in Lemma~\ref{lem2}. From~\eqref{eq21}, we have
\begin{equation}\label{eq63}
\begin{split}
\abs{I_{3,p}}
& \leq \sum_{l=\delta_p +1}^{\infty}
\sum_{m=1}^{\delta_p} l\abs{l-m} (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m} \\
& \leq \sum_{l=\delta_p + 1}^{\infty} l^2 (c_l^{(p)})^2 \abs{z}^{2l}
\cdot \sum_{m=1}^{\delta_p} (c_m^{(p)})^2 \abs{z}^{2m},
\end{split}
\end{equation}
as $l\abs{l-m} =l(l-m) <l^2$.

From~\eqref{eq48}, we have
\begin{equation}\label{eq65}
\begin{split}
\sum_{l=\delta_p+1}^{\infty} l^2 (c_l^{(p)})^2 \abs{z}^{2l}
& \leq \sum_{l=\delta_p +1}^{\infty} l^2 C p
\frac{1}{(2r)^{2l}} \frac{1}{ \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^p } \abs{z}^{2l } \\
& = C p \frac{1}{ \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^p }
\sum_{l=\delta_p +1}^{\infty} l^2 \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^{2l}.
\end{split}
\end{equation}
By a simple computation, we get for $0\leq \xi <1$,
\begin{equation*}
\begin{split}
\sum_{q=N+1}^{\infty} q(q+1) \xi^{q-1}
& = \parens*{\sum_{q=N+1}^{\infty} \xi^{q+1}}'' 		\\
& = \parens[\Bigg]{\frac{ \xi^{N+2}}{ 1 -\xi }}'' 	\\
& = \frac{ (N+2)(N+1) \xi^N - 2N(N+2)\xi^{N+1} + N(N+1) \xi^{N+2}}{ (1-\xi)^3 } \\
& \leq \frac{(N+2) (N+1) \xi^N } { (1-\xi)^3 }.
\end{split}
\end{equation*}
Thus,
\begin{equation}\label{eq66}
\begin{split}
\sum_{l=\delta_p +1}^{\infty}l^2 \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^{2l}
& \leq \parens[\Bigg]{\frac{\abs{z}}{2r}}^2 \sum_{l=\delta_p +1}^{\infty} l(l+1) \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^{2(l-1)} \\
& \leq \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2 (\delta_p+2) (\delta_p + 1)
\frac{ \parens[\Big]{\frac{\abs{z}}{2r}}^{2\delta_p}} {\parens[\Big]{1 - \parens[\Big]{\frac{\abs{z}}{2r}}^2}^3 } \\
& \leq \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2 (\delta_p + 2) (\delta_p + 1) \cdot \parens[\Bigg]{\frac{4}{3}}^3
\parens[\Bigg]{\frac{\abs{z}}{ 2r }}^{2 \delta_p }.
\end{split}
\end{equation}
Therefore, by~\eqref{eq65} and~\eqref{eq66}, we get
\begin{equation}\label{eq67}
\sum_{l=\delta_p+1}^{\infty} l^2 (c_l^{(p)})^2 \abs{z}^{2l}
\leq C \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2 (\delta_p + 2) (\delta_p + 1) \parens[\Bigg]{\frac{4}{3}}^3 \cdot p \frac{1}{ \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^p }
\parens[\Bigg]{\frac{\abs{z}}{ 2r }}^{2 \delta_p }.
\end{equation}
By~\eqref{eq22}, \eqref{eq52} for $\tau=1$ and~\eqref{eq67}, we get
\begin{equation}\label{eq68}
\begin{split}
\sum_{l=\delta_p+1}^{\infty} l^2 (c_l^{(p)})^2 \abs{z}^{2l}
& \leq C \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2 (\delta_p + 2) (\delta_p + 1) \parens[\Bigg]{\frac{4}{3}}^3 \cdot C^2
\frac{1}{2^{2\alpha' p}} \beta_p^{\mathbb{D}^*}(z) \\
& \leq C' \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2 (p+2) (p + 1)
\frac{1}{2^{2\alpha' p}} \beta_p^{\mathbb{D}^*}(z).
\end{split}
\end{equation}
From~\eqref{eq16}, \eqref{eq63} and~\eqref{eq68}, we get
\begin{equation}\label{eq70}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{ I_{3,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }}
\leq
\parens[\big]{\log \abs{z}^2}^2 \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2
\bracks[\Bigg]{C' (p+2)(p+1) \frac{1}{2^{ 2\alpha' p}}}.
\end{equation}
Since the function $f(x) = x^2(\log x^2)^2$ is bounded on the interval $\mathopen{]}0,1\mathclose{[}$, from~\eqref{eq70}, we get
\begin{equation}\label{eq71}
\sup_{ 0<\abs{z} \leq c' p^{-A'}}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{3,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }}
\leq C_2 \, p^2 \frac{1}{2^{2\alpha' p}}.
\end{equation}
Since $c' p^{-A'} > b e^{-p^{\gamma}}$ for $p\gg 1$, we get~\eqref{eq26} for $j=3$ from~\eqref{eq71}.
\let\qed\relax
\end{proof}

\begin{proof}[Proof of Lemma~\ref{lem}(\ref{lem_1}) for $\boldsymbol{j=4}$]
We assume $\abs{z}\leq c' p^{-A'}$, $p \gg 1$ as in Lemma~\ref{lem2}. By~\eqref{eq21}, we have
\begin{equation}\label{eq72}
\begin{split}
\abs{I_{4,p}}
& \leq \sum_{l=\delta_p+1}^{\infty} \sum_{m=\delta_p+1}^{\infty} l \abs{l-m} (c_l^{(p)})^2 (c_m^{(p)})^2 \abs{z}^{2l+2m} \\
& \leq \sum_{l=\delta_p + 1}^{\infty} l^2 (c_l^{(p)})^2 \abs{z}^{2l}
\cdot \sum_{m=\delta_p + 1 }^{ \infty} (c_m^{(p)})^2 \abs{z}^{2m} \\
& \quad + \sum_{l=\delta_p +1}^{\infty} l (c_l^{(p)})^2 \abs{z}^{2l}
\cdot \sum_{m=\delta_p + 1}^{\infty} m (c_m^{(p)})^2 \abs{z}^{2m}.
\end{split}
\end{equation}
From~\eqref{eq48}, as in~\eqref{eq58}, we get
\begin{equation}\label{eq51}
\begin{split}
\sum_{m=\delta_p+1}^{\infty} (c_m^{(p)})^2 \abs{z}^{2m}
& \leq \sum_{m=\delta_p +1}^{\infty} C p \frac{1}{(2r)^{2m}}
\frac{1}{ \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^p } \abs{z}^{2m} \\
& \leq Cp \frac{1}{ \abs[\big]{\log\parens[\big]{\abs{2r}^2}}^p } 4\cdot \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^{2\delta_p +2}.
\end{split}
\end{equation}
From~\eqref{eq52} for $\tau =1$ and~\eqref{eq51}, we get
\begin{equation}\label{eq53}
\sum_{m=\delta_p+1}^{\infty} (c_m^{(p)})^2 \abs{z}^{2m} \leq 4C \frac{1}{2^{ 2\alpha' p}} \beta_p^{\mathbb{D}^*}(z) \parens[\Bigg]{\frac{\abs{z}}{ 2r }}^2.
\end{equation}
From~\eqref{eq58}, \eqref{eq68}, \eqref{eq72} and~\eqref{eq53}, we get
\begin{equation}\label{eq73}
\begin{split}
\abs{I_{4,p}} 
& \leq C' \parens[\Bigg]{\frac{\abs{z}}{2r}}^2 (p+2)(p+1)
\frac{1}{2^{2\alpha' p }} \beta_p^{\mathbb{D}^*}(z) \cdot 4C \frac{1}{2^{2\alpha' p}} \beta_p^{\mathbb{D}^*}(z)
\parens[\Bigg]{\frac{\abs{z}}{2r}}^2 \\
& \quad + \bracks*{\parens[\Bigg]{\frac{4}{3}}^2 C' \parens[\Bigg]{\frac{\abs{z}}{2r}}^2 (p + 1) \frac{1}{2^{2\alpha' p}}
\beta_p^{\mathbb{D}^*}(z)}^2 \\
& \leq C'' p^2 \frac{1}{2^{4\alpha' p}}
\parens[\Bigg]{\frac{\abs{z}}{2r}}^4 \beta_p^{\mathbb{D}^*}(z)^2.
\end{split}
\end{equation}
Thus, by~\eqref{eq73} we obtain
\begin{equation}\label{eq74}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{4,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }}
\leq C'' p^2 \frac{1}{2^{4\alpha' p}}
\parens[\Bigg]{\frac{\abs{z}}{2r}}^4 \parens[\big]{\log \abs{z}^2}^2.
\end{equation}
Since the function $f(x) = x^4 (\log x^2)^2$ is bounded on the interval $\mathopen{]}0,1\mathclose{[}$, from~\eqref{eq74}, we get
\begin{equation}\label{eq75}
\sup_{ 0<\abs{z} \leq c' p^{-A'}}
\abs*{\parens[\big]{\log \abs{z}^2}^2 \cdot \frac{I_{4,p}}{\parens[\big]{\beta_p^{\mathbb{D}^*}(z)}^2 }}
\leq C''' p^2 \frac{1}{2^{4\alpha' p}}.
\end{equation}
Since $c' p^{-A'} > b e^{-p^{\gamma}}$ for $p\gg 1$, we get~\eqref{eq26} for $j=4$ from~\eqref{eq75}.
\let\qed\relax
\end{proof}

The proof of Lemma~\ref{lem} is completed.

\subsection{Proof of Corollary~\ref{cor}}\label{s2.3}

Let $\Gamma$ be a geometrically finite Fuchsian group of the first kind, without elliptic elements, then as explained in~\cite[p.~956]{AMM21}, $\Sigma \coloneqq \Gamma \backslash \mathbb{H}$ can be compactified by finitely many points $D=\{a_1,\dotsc, a_N\}$ into a compact Riemann surface $\bar{\Sigma}$ with genus $g$ such that the following equivalent conditions~\eqref{cond_1}--\eqref{cond_4} are fulfilled:
\begin{enumerate}\romanenumi
\item \label{cond_1} $\Sigma = \bar{\Sigma} \smallsetminus D$ admits a complete KÃ¤hler--Einstein metric $\omega_{\Sigma}$ with $\Ric_{\omega_{\Sigma}} = - \omega_{\Sigma}$;
\item \label{cond_2} $2g-2+N >0$;
\item \label{cond_3} the universal cover of $\Sigma$ is the upper-half plane $\mathbb{H}$;
\item \label{cond_4} $L=K_{\bar{\Sigma}} \otimes
\mathcal{O}_{\bar{\Sigma}}(D)$ is ample, where $K_{\bar{\Sigma}} = T^{*(1,0)} \bar{\Sigma}$ is the canonical line bundle on $\bar{\Sigma}$.
\end{enumerate}

From~\cite[Lemma~6.2]{AMM21}, there exists a singular Hermitian metric on $L$ such that $(\Sigma, \omega_{\Sigma})$ and the formal square root of $(L,h)$ satisfy the conditions~\eqref{alpha} and~\eqref{beta}.

From~\cite[the proof of Lemma~6.2]{AMM21}, $\omega_{\Sigma}$ is the KÃ¤hler--Einstein form of constant negative curvature $-4$, induced by the Poincar\'e form $\omega_{\mathbb{H}} = \frac{ i\,\dd z \wedge \dd \bar{z}}{ 4\abs{\Imag z}^2}$ on $\mathbb{H}$, and every $a\in D$ has a coordinate neighborhood $(\bar{U}_a, z)$ in $\bar{\Sigma}$ such that in this coordinate $\omega_{\Sigma}$ is exactly given by $\omega_{\mathbb{D}^*}(z)$ on $U_a = \bar{U}_a \smallsetminus \{a\}$. Moreover, the curvature of the line bundle $\parens[\big]{\restr{L}{\Sigma}, h}$ is given by $-2i\omega_{\Sigma}$ and thus the curvature of the (formal) square root of $(L,h)$ is $-i\omega_{\Sigma}$. Therefore, from~\cite[Corollary~2.4]{AMM21}, we get that for any $k,m\in \mathbb{N}$ and any compact set $K\subset \Sigma$,
\begin{equation}\label{eq76}
B_p(x) = \frac{1}{\pi} p - \frac{1}{2\pi} + \mathcal{O}(p^{-k}), \quad
\text{in $C^m(K)$ as $p\rightarrow +\infty$}.
\end{equation}
Hence, from~\eqref{eq6}, we get
\begin{equation}\label{eq77}
\sup_{ K } \abs*{\frac{ J^*_{p,(2)} \omega_{\FS,p}(z) } { 2p \, \omega_{\Sigma} (z) } - \frac{1}{2\pi}} = \mathcal{O}(p^{-\infty}).
\end{equation}
On the coordinate neighborhood $U_a$ of $a\in D$, we have $\omega_{\Sigma} = \omega_{\mathbb{D}^*}$. Combine Theorem~\ref{thm_MainThm} and~\eqref{eq77}, we get
\begin{equation}\label{eq79}
\sup_{ \Sigma } \abs*{\frac{ J^*_{p,(2)} \omega_{\FS,p}(z) } { p \, \omega_{\Sigma} (z) }} = \mathcal{O}(p^3).
\end{equation}

As explained in~\cite[(6.36)]{AMM21}, the space $\mathcal{S}_{2p}^{\Gamma}$ of cusp forms (Spitzenformen) of weight $2p$ of~$\Gamma$ endowed with the Petersson scalar product is isometric with the space of ${\bL}^2$ holomorphic sections of $L^p$ over $\Sigma$ (cf.~\eqref{eq4}) with respect to the volume form $\omega_{\Sigma}$ and the metric $h^p$ on $L^p$. The Bergman kernel $B_p$ of $H^0_{(2)}(\Sigma, L^p)$ can be identified to the Bergman kernel $S_p^{\Gamma}$ of $\mathcal{S}_{2p}^{\Gamma}$ (cf.~\cite[(6.37)]{AMM21}). Thus, by~\eqref{eq6} and~\eqref{eq.berg}, we get
\begin{equation}\label{eq.berg1}
\omega_{\Sigma}^{\Ber,p} = \frac{i}{2\pi} \partial \bar{\partial} \log \parens[\big]{B_p(z)},
\qquad
\frac{1}{p} J^*_{p,(2)} \omega_{\FS,p} = \frac{1}{\pi} \omega_{\Sigma} + \frac{1}{ p} \omega_{\Sigma}^{\Ber,p}.
\end{equation}
From~\eqref{eq79} and~\eqref{eq.berg1}, we get~\eqref{eq10}. The proof of Corollary~\ref{cor} is completed.

\section*{Declaration of interests}

The authors do not work for, advise, own shares in, or receive funds from any organization that could benefit from this article, and have declared no affiliations other than their research organizations.

\newpage

\printbibliography

\end{document}